# Validity of analytical solutions for the groundwater response to Earth tides

Jose Manuel Bastias<sup>1</sup>, Gabriel Rau<sup>2</sup>, and Philipp Blum<sup>3</sup>

<sup>1</sup>Karlsruhe Institute of Technology <sup>2</sup>The University of Newcastle <sup>3</sup>Karlsruhe Institute of Technology (KIT)

November 22, 2022

#### Abstract

Harmonic Earth tide components in well water levels have been used to estimate hydraulic and geomechanical subsurface properties. However, the validity of various methods based on analytical solutions has not been established. First, we review the theory and examine the latest analytical solution used to relate well water levels to Earth tides. Second, we develop and verify a novel numerical model coupling hydraulics and geomechanics to Earth tide strains. Third, we assess subsurface conditions over depth for a range of realistic properties. Fourth, we simulate the well water level response to Earth tide strains within a 2D poroelastic layered aquifer system confined by a 100 m thick aquitard. We find that the analytical solution matches two observations (amplitudes and phases) to multiple unknown parameters leading to non-unique results. We reveal that undrained and confined conditions are necessary for the validity of the analytical solution. This occurs for the dominant M\_2 frequency at depths >50 m and requires specific storage at constant strain of  $S_{\epsilon}$  [?] 10<sup>-6</sup> m<sup>-1</sup>, in combination with hydraulic conductivity of the aquitard  $k_1$  [?] 5\*10<sup>-5</sup> ms<sup>-1</sup> and aquifer ka [?] 10<sup>-4</sup> ms<sup>-1</sup>. We further illustrate that the analytical solution is valid in unconsolidated systems, whereas consolidated systems require additional consideration of the Biot modulus. Overall, a priori knowledge of the subsurface system supports interpretation of the groundwater response. Our results improve understanding of the effect of Earth tides on groundwater systems and its interpretation for subsurface properties.

# Validity of analytical solutions for the groundwater response to Earth tides

Jose M. Bastias Espejo<sup>1</sup>, Gabriel C. Rau<sup>1,2</sup>, Philipp Blum<sup>1</sup>

<sup>1</sup>Institute of Applied Geosciences (AGW), Karlsruhe Institute of Technology (KIT), Karlsruhe, Germany <sup>2</sup>School of Environmental and Life Sciences, The University of Newcastle, Newcastle, Australia

## 6 Key Points:

1

2

3

4 5

12

7	•	We develop and verify an advanced simulator for the well water level response to
8		Earth tides
9	•	Subsurface property estimation requires undrained and confined conditions occur-
10		ring at depths $> 50 \text{ m}$
11	•	Amplitudes and phases from numerical and analytical solutions systematically di-

verge reflecting theory simplifications

Corresponding author: Jose M. Bastias Espejo, jose.bastias@kit.edu

#### 13 Abstract

Harmonic Earth tide components in well water levels have been used to estimate hydraulic 14 and geomechanical subsurface properties. However, the validity of various methods based 15 on analytical solutions has not been established. First, we review the theory and exam-16 ine the latest analytical solution used to relate well water levels to Earth tides. Second, 17 we develop and verify a novel numerical model coupling hydraulics and geomechanics 18 to Earth tide strains. Third, we assess subsurface conditions over depth for a range of 19 realistic properties. Fourth, we simulate the well water level response to Earth tide strains 20 within a 2D poroelastic layered aquifer system confined by a 100 m thick aquitard. We 21 find that the analytical solution matches two observations (amplitudes and phases) to 22 multiple unknown parameters leading to non-unique results. We reveal that undrained 23 and confined conditions are necessary for the validity of the analytical solution. This oc-24 curs for the dominant  $M_2$  frequency at depths > 50 m and requires specific storage at 25 constant strain of  $S_{\epsilon} \geq 10^{-6} m^{-1}$ , hydraulic conductivity of the aquitard of  $k_l \leq 5 \cdot 10^{-5} ms^{-1}$  and aquifer of  $k_a \geq 10^{-4} ms^{-1}$ . We further illustrate that the analytical 26 27 solution is valid in unconsolidated systems, whereas consolidated systems require addi-28 tional consideration of the Biot modulus. Overall, a priori knowledge of the subsurface 29 system supports interpretation of the groundwater response. Our results improve un-30 derstanding of the effect of Earth tides on groundwater systems and its interpretation 31 for subsurface properties. 32

<sup>33</sup> Plain Language Summary

Earth tide induced strains in the subsurface lead to well water level fluctuations 34 in groundwater monitoring wells. This groundwater response has been interpreted with 35 analytical solutions to estimate aquifer properties. However, analytical solutions are based 36 on simplified assumptions whose conditions of validity have not yet been tested. We de-37 velop a new approach to simulate the influence of Earth tides on groundwater based on 38 fundamental physical principles. We simulate realistic conditions and compare our re-39 sults to those from analytical solution to determine the hydraulic and subsurface con-40 ditions under which simplified interpretations are valid. Our results improve understand-41 ing of the effects of Earth tides on groundwater systems and interpretation of subsur-42 face properties from well water levels. 43

## 44 1 Introduction

Earth tides have long been observed to influence groundwater systems (Meinzer, 45 1939), a phenomenon that is commonly expressed as harmonic water level fluctuations 46 in monitoring wells (Merritt, 2004). Analytical solutions based on simplified concepts 47 have been developed to enable calculation of subsurface hydraulic and geomechanical prop-48 erties (Cutillo & Bredehoeft, 2011; Wang & Manga, 2021; Rau, Cuthbert, Acworth, & 49 Blum, 2020). Since Earth tides are a ubiquitous natural force, their response should be 50 contained in the data from numerous wells around the world. In fact, a recent review 51 found that interpreting the groundwater response to Earth tides is underutilized and that 52 further development offers the potential for widespread application, which in turn would 53 lead to increased knowledge of the subsurface (McMillan et al., 2019). 54

Estimating subsurface hydro-geomechanical properties requires good quality monitoring data of groundwater and Earth tides (Merritt, 2004; Rau, Cuthbert, Post, et al., 2020). Of main interest are the amplitude ratio and phase shift between the Earth tide strain and its well water level response. The first step is to extract the subtle harmonic components whose frequencies are well defined (Rojstaczer, 1988; Rojstaczer & Riley, 1990). Various approaches have been used to identify and extract the dominant amplitudes and phases (Turnadge et al., 2019). Recent research illustrated that least-squares fitting of the dominant harmonic tidal components to measurements provides the most
 robust results (Schweizer et al., 2021).

Early research developed and tested an analytical solution to estimate subsurface 64 properties from the relationship between strain and water levels in a fully confined aquifer 65 (Cooper Jr et al., 1967; Hsieh et al., 1987). This was extended to include leaky condi-66 tions and allow concurrent use of multiple frequencies (Rojstaczer, 1988; Rojstaczer & 67 Riley, 1990). Rojstaczer and Agnew (1989) studied the dependency of porosity and elas-68 tic parameters to a real deformation of a poro elastic medium. High sensitivity was re-69 70 ported when the applied strains occurs in low porosity and the increase with decreasing compressibility (inverse of the bulk modulus) of the solid matrix. 71

Field measurements of the groundwater response to Earth tides has resulted in neg-72 ative and positive phase shifts between strain and well water levels. Negative phase shifts 73 are interpreted as fully confined conditions and horizontal flow only (Hsieh et al., 1987). 74 For example, hydraulic conductivity and specific storage were estimated from negative 75 phase shifts (Xue et al., 2016; Allègre et al., 2016). However, positive phase shifts in the 76 field were also observed and attributed to vertical flow through leaky aquitards (Roeloffs 77 et al., 1989). This was interpreted using the analytical solution of vertical flow in an ho-78 mogeneous aquifer caused by a harmonic load or stress (Wang, 2017). 79

Recent research included modifications to the original analytical solution by Hsieh 80 et al. (1987) to account for more realistic conditions. Most notable is the work by Wang 81 et al. (2018) who developed an extended analytical solution which includes vertical leak-82 age to model a two-layered aquifer system. Gao et al. (2020) investigated the well skin 83 effect which originates from the fact that the formation around a well is disturbed and 84 well water storage on the well water level response to Earth tides. The authors found 85 that the skin effect may significantly delay the well water level phase response to Earth 86 tides. In addition, Guo et al. (2021) developed a model for tidal response with a fault 87 passing through the aquifer based on a fault-guided fracture network to estimate frac-88 ture properties. They found that the hydraulic diffusivity in the fault damage zone higher 89 than previously established values, but also that it remains below estimates based on in-90 duced seismicity migration. Sun et al. (2020) reviewed four of the most common ana-91 lytical models to estimate hydraulic properties with Earth tidal analysis. They estimated 92 hydraulic properties from a real data set and provide a range of applicability of the dif-03 ferent models based on the transmissivity of the aquifer. 94

While analytical models offer a convenient approach to estimate hydraulic properties, their applicability is limited through simplifying assumptions arising from fundamental physics, conceptual model or boundary conditions. These include for example only radial flow, negligible horizontal displacement of the aquifer, confined conditions and undrained conditions, unconsolidated materials and no gravity. But it has been reported that some of these assumptions may significantly influence the estimated subsurface properties (Wang et al., 2019; Zhu & Wang, 2020).

Numerical modeling of tidal effects is common in coastal (Abarca et al., 2013; Zhang 102 et al., 2021) and adjacent settings (Jardani et al., 2012; Pendiuk et al., 2020; Alcaraz et 103 al., 2021), likely because the loading effect of ocean tides is much larger than that caused 104 solely by Earth tide forces. So far, modeling the subsurface response to ocean tides has 105 considered poro elastic conditions and harmonic loading by the weight of the water, there-106 fore, solved as a consolidation problem. In contrast, the pressurization of an inland aquifer 107 is produced by changes in the pore space volume of the porous material due to strains 108 (i.e., eigenstrains) caused by the gravitational influence from the movement of celestial 109 bodies. Moreover, the change of the confined pore pressure is generally measured using 110 an observation well which causes fluid movement. 111

To the best of our knowledge, only hydraulic modeling approaches neglecting any 112 geomechanical effects, i.e., groundwater flow without coupled poro elasticity, have so far 113 been used to investigate the groundwater response to Earth tides. For example, Wang 114 et al. (2019) simulated the effect of capillarity of the unsaturated zone in one dimension. 115 They found out that the assumption of fixed water table can lead to erroneous estima-116 tion of subsurface properties. Zhu and Wang (2020) simulated a two layered system to 117 study the effect of leakage through an aquitard and concluded that simplifications in the 118 analytical model lead to overly conservative estimates of vertical flux between layers. Wang 119 and Manga (2021) provide a summary of these works. 120

The confined pore pressure generated as result of Earth tide strains is a mechan-121 ical phenomenon caused by the elastic deformation of the porous matrix. Furthermore, 122 unlike for traditional hydraulic testing approaches, there is a general lack of work inves-123 tigating the effect of realistic conditions and assumptions on interpretations using an-124 alytical solutions. Investigating the influence of limiting assumptions and realistic sub-125 surface conditions to better understand the applicability of analytical solutions requires 126 development of more advance numerical models that also consider coupling with geome-127 chanics. 128

The objetive of this study is therefore to (1) develop a numerical model for the ground-129 water responses to Earth tides, which couples hydraulic and geomechanical processes, 130 (2) critically examine assumptions upon which analytical solutions are based, (3) inves-131 tigate and compare response conditions as well as the influence of geomechanical prop-132 erties. Thus, our work significantly improves our understanding of the coupled physics, 133 which controls the well response in a poro elastic medium. These results can act as a prac-134 tical guide for improved estimation of aquifer properties due to the groundwater response 135 to Earth tides. 136

#### 137 2 Methodology

#### 138

#### 2.1 Fundamental theory of the groundwater response to Earth tides

Earth tides are displacements of the solid Earth's crust caused by the gravitational 139 forces of celestial bodies that move in relation to the Earth. Such displacements are typ-140 ically expressed as harmonic signals that can be predicted from well-known astronom-141 ical relationships. Earth tide forces are dominant at distinct frequencies within the semi-142 diurnal and diurnal range, e.g.,  $M_2$  at 1.97322 cycles per day (cpd) or  $S_1$  at 1.0 cpd. Un-143 der tidal forcing, the pore volume of the subsurface elastically deforms depending on the 144 mechanical properties of the filling material resulting in a small change of volume. If the 145 subsurface is saturated, the filling fluid has to adapt to the new available pore space which 146 raises or lowers the pore pressure. The processes can be mathematically represented by 147 the *Biot* consolidation theory. 148

Biot (1941) developed the constitutive laws which relate the applied forces (stresses) 149 with deformation (strains) and motion within a compressible porous medium. For the 150 purpose of modeling, these laws are formulated in the form of mathematical equations 151 which consist of four basic variables stress  $(\sigma_{ij})$ , strain  $(\epsilon_{ij})$ , pore pressure  $(p_f)$  and in-152 cremental of fluid content  $(\xi)$ . The mechanical variables (stress or strain) can be related 153 with one of the fluid quantity (pore pressure or fluid content) to form independent vari-154 able and therefore mathematical equations. For the particular case of Earth tides, is con-155 venient to express the pore elastic equations in terms of stress and pore pressure, also 156 termed pure stiffness formulation 157

$$\sigma_{ij} = \left(K_u - \frac{2G}{3}\right)\delta_{ij}\epsilon + 2G\epsilon_{ij} - \alpha M\delta_{ij}\xi \tag{1}$$

158 and

$$p_f = M(-\alpha\epsilon + \xi),\tag{2}$$

where  $\alpha$  is the Biot poroelastic coefficient,  $K_u$  is the undrained bulk modulus, G is the shear modulus,  $\epsilon$  is the volumetric strain ( $\epsilon = \epsilon_{xx} + \epsilon_{yy} + \epsilon_{zz}$ ),  $\delta_{ij}$  is the Kronecker delta and M is the Biot modulus which is defined as,

$$\frac{1}{M} = \frac{n}{K_f} + \frac{(1-\alpha)(\alpha-n)}{K} = \frac{\alpha}{BK_u},\tag{3}$$

where n is the porosity,  $K_f$  is the bulk modulus of the fluid, K is the drained bulk modulus (related to the undrained bulk modulus as  $K = K_u - \alpha^2 M$ ) and B is the Skempton coefficient. Fluid transport is modeled with the fluid balance equation as

$$S_{\epsilon} \left( \frac{\partial p}{\partial t} + BK_u \frac{\partial \epsilon}{\partial t} \right) = k_{ij} \nabla^2 p_f + Q \tag{4}$$

where  $k_{ij}$  is the porous medium permeability tensor, Q represents sinks or sources,  $\epsilon_{ij} = \nabla u_i$  relates strain with displacement, often prefered in simulation codes (Verruijt, 2013; Kolditz et al., 2012; Flemisch et al., 2011; Keilegavlen et al., 2021), and  $S_{\epsilon}$  is the specific storage at constant strain.  $S_{\epsilon}$  is related to the Biot modulus as

$$S_{\epsilon} = \frac{1}{M},\tag{5}$$

Equations 1, 2 and 4 can be solved in a coupled manner with appropriate boundary conditions and represent the elastic deformation and fluid movement in a porous medium.

#### 171 2.2 Analytical solution

<sup>172</sup> When uniaxial-vertical strain and zero incremental vertical stress are assumed (this <sup>173</sup> occurs only when one of the principal stresses is non-zero and the stress does not change <sup>174</sup> with depth), the subsurface is mechanically restricted to move only in the vertical di-<sup>175</sup> rection, e.g., land surface subsidence due to consolidation occurs primarily in the ver-<sup>176</sup> tical direction (Herrera-García et al., 2021). Under such conditions  $\epsilon_{xx} = \epsilon_{yy} = 0$  and <sup>177</sup>  $\sigma_{zz} = 0$  which leads to a simplified version of equations 1 and 2 as

$$\sigma_{zz} = 0 = \left(K + \frac{4G}{3}\right)\epsilon_{zz} - \alpha p_f \tag{6}$$

178 and

$$p_f = M(-\alpha \epsilon_{zz} + \xi). \tag{7}$$

179 Combining 6 and 7 to eliminate  $\epsilon_{zz}$  gives

$$\xi = Sp_f,\tag{8}$$

180 where

$$S = \frac{1}{M} + \frac{3\alpha^2}{3K + 4G}.\tag{9}$$

This is the definition of storage coefficient in hydrology (Cheng, 2016; Verruijt, 2013; Wang, 2017). The specific storage,  $S_s$ , is obtained when the specific weight of the fluid is considered as

$$S_s = S\rho_f g,\tag{10}$$

where  $\rho_f$  is the density of the filling fluid and g Earth's gravitational acceleration constant.

With this derivation, we stress the conceptual difference between the specific storage at constant strain (equation 5) and the storage coefficient with uniaxial strain (equation 9). Please note that S approaches  $S_{\epsilon}$  when  $K >> K_f$ , such as when grains are less compressible than water, e.g. sand or gravel (Verruijt & Van Baars, 2007; Lambe & Whitman, 1991; Freeze & Cherry, 1979). <sup>191</sup> Hydraulic head can be used as a proxy for pore pressure in equation 4,  $h = p\rho_f^{-1}g^{-1}$ . <sup>192</sup> Moreover, in a confined aquifer with a constant thickness  $H_a$ , hydraulic conductivity can <sup>193</sup> be express in terms of transmissivity ( $T = k_a H_a$ ) and specific storage at constant strain <sup>194</sup> in terms of storativity at constant strain  $S_{\epsilon,t} = S_{\epsilon}H_a$  as (Cheng, 2016; Verruijt, 2013; <sup>195</sup> Wang, 2017),

$$S_{\epsilon,t}\left(\frac{\partial h}{\partial t} + \frac{BK_u}{\rho_f g}\frac{\partial \epsilon}{\partial t}\right) = T\left[\frac{\partial^2 h}{\partial x^2} + \frac{\partial^2 h}{\partial y^2} + \frac{\partial^2 h}{\partial z^2}\right] + Q.$$
 (11)

For practical reasons, equation 11 can be reformulated into cylindrical coordinates assuming only radial flow (C. Jacob & Lohman, 1952; C. E. Jacob, 1946), also vertical leakage con be included in the sink/source in terms of hydraulic conductivity of the layer on top  $(k_l)$  and thickness of such layer  $(H_l)$  expecting  $k_a k_l^{-1} \gg 1$  as  $Q = k_l h H_l^{-1}$ ,

$$S_{\epsilon,t}\left(\frac{\partial h}{\partial t} + \frac{BK_u}{\rho_f g}\frac{\partial \epsilon}{\partial t}\right) = T\left[\frac{\partial^2 h}{\partial r^2} + \frac{1}{r}\frac{\partial h}{\partial r}\right] + \frac{k_l}{H_l}h.$$
(12)

In this work we use the terms aquifer and aquitard to reflect layers of higher and 200 lower hydraulic conductivity, respectively, as is consistent with the terminology used in 201 previous works. Equation 12 was used to derive analytical solutions that describe the 202 groundwater response to Earth tides in a fully confined aquifer (Hsieh et al., 1987) and 203 in an aquifer with vertical leakage (Wang et al., 2018). A detailed derivation of the more 204 versatile leaky solution is presented in Wang et al. (2018), but is also included in Ap-205 pendix A of this work. The solution describes the water level in a well  $(h_w)$  in terms of 206 the hydraulic conductivity of the aquifer  $(k_a)$ , hydraulic conductivity of the aquitard  $(k_l)$ 207 and the specific storage at constant strain of the aquifer  $(S_{\epsilon})$ . Fluid level changes in the 208 well are caused by forces generated at the far field (far away from the radius of influence 209 of the observation well). Assuming undrained conditions ( $\xi = 0$ , in equation 2) and as-210 suming  $\alpha = 1$  representing unconsolidated systems such changes can be quantify as 211

$$p = -M\epsilon_G,\tag{13}$$

where  $\epsilon_G$  is a external gravitational strain. The change of water level in a well due to an areal strain is graphically shown in Figure 1a and 1b for two given times from  $t_0 =$ 0 to t = t with gravitational strains from  $\epsilon_G = 0$  to  $\epsilon_G = \epsilon_G(t)$ .

The amplitude ratio and phase shift between the piezometric head at a distance from the well beyond its radius of influence and the water level in the well are expressed as (Hsieh et al., 1987),

$$A = \left| h_w / \frac{BK_u \epsilon_0}{\rho g} \right| = \left| h_w \alpha / \frac{M\epsilon_0}{\rho g} \right| \tag{14}$$

218 and

$$\Delta \phi = \arg\left(h_w / \frac{BK_u \epsilon_0}{\rho g}\right) = \arg\left(h_w \alpha / \frac{M\epsilon_0}{\rho g}\right). \tag{15}$$

Here,  $\epsilon_0$  is the amplitude of the volumetric strain signal, which can be obtained from software based on tidal catalogs, e.g. PyGtide (Rau, 2018), ETERNA PREDICT (Wenzel, 1996), *TSoft* (Van Camp & Vauterin, 2005). The analytical solution is subject to the assumptions under which it was derived: (1) undrained conditions, (2) the confining layer has negligible specific storage, (3) the flow is horizontal in the aquifer, (4) the well is represented by a line with length matching the aquifer extent, (5) the deformation is only vertical, (6) no gravity.

Equations 14 and 15 can be inverted to estimate constant values of equation 12 using any approach suitable for non-linear algorithm estimation (gradient methods), e.g. least squares (Rau, Cuthbert, Acworth, & Blum, 2020). In fact, this approach has been used to estimate aquifer hydraulic conductivity, specific storage and aquitard leakage from the amplitude and phase response of groundwater to Earth tides (Rau, Cuthbert, Acworth, & Blum, 2020). However, estimating three parameters from two inputs leaves the
problem under-determined and leads to non-unique estimates. We note that neither the
implications of the assumptions nor the non-uniqueness problem have been investigated
for practical use of this solution.

Gradient methods such as the *Levenburg-Marquardt* often use in least-squares to numerically search for the nearest (local) minimum to the given initial condition and are readily implemented in SciPy (Virtanen et al., 2020). The obtained solution from the function fitting model finds the local minimum and depends on the initial guess. To investigate the impact of non-uniqueness by the initial conditions on the parameter estimation of Wang et al. (2018), we systematically explore the solution space of a fitting function. The fitting function was formulated as

$$FO_{Amplitude} = A_{obs} - A(k_a, S_{\epsilon}, k_l) \tag{16}$$

242 and

262

$$FO_{Phase} = \phi_{obs} - \Delta \phi(k_a, S_\epsilon, k_l) \tag{17}$$

where  $A(k_a, S_{\epsilon}, k_l)$  and  $\Delta \phi(k_a, S_{\epsilon}, k_l)$  are equation 14 and 15. The aquifer thickness and aquitard depth were arbitrarily defined to be 1 *m* and 100 *m*, respectively; the radius of the well to 0.2 m and Earth tide frequency of 1.93 cpd;  $A_{obs}$  and  $\phi_{obs}$  are objective amplitude ratio and phase shift values given by the user.

The least-squares solver minimizes the difference between  $A_{obs}-A(k_a, S_{\epsilon}, k_l)$  and 247  $\phi_{obs} - \Delta \phi(k_a, S_{\epsilon}, k_l)$  of equations 16 and 17 by iterating through a combination of values 248 of  $k_a$ ,  $S_{\epsilon}$  and  $k_l$ . An array consisting of discrete values within realistic ranges for am-249 plitude ratio (0.001 < A < 1) and phase ( $-90^{\circ} < \Delta \phi < 90^{\circ}$ ) where generated. For 250 each pair of amplitude ratio and phase shift, the solution space was solved using least-251 squares of SciPy. The tolerance for termination by the change of the cost function was 252 set to be  $3 \cdot 10^{-6}$  and units for 16 and 17 where set to days so as to increase the numeric 253 values and avoid errors. The sensitivity of the method was tested by generating 1000 sam-254 ples of  $k_a$ ,  $S_{\epsilon}$  and  $k_l$  generated by a random log uniform distribution ranging from  $1 \cdot 10^{-7} \leq k_a \leq 1 \cdot 10^{-3} m s^{-1}$ ,  $1 \cdot 10^{-7} \leq S_{\epsilon} \leq 1 \cdot 10^{-5} m^{-1}$  and  $1 \cdot 10^{-8} \leq k_l \leq 1 \cdot 10^{-4}$ 255 256  $ms^{-1}$ . Each trio of samples was set as initial condition in equation 16 and 17. Starting 257 from different initial values allows the solver to find potentially different local minima. 258 The outputs were stored and the maximal difference between each estimated property 259 was used as a proxy for non-uniqueness. This approach allows identification of values 260 within the solution space where non-uniqueness is an issue. 261

#### 2.3 Numerical model

The generic equations presented in Section 2.1 can be solved analytically for spe-263 cific boundary conditions (Hsieh et al., 1987) but have not been solved numerically in 264 the context of Earth tides. Here, we develop a novel numerical approach for simulating 265 the groundwater response to Earth tides. This allows a more realistic physical represen-266 tation compared to the limiting assumptions of the analytical solution presented in Sec-267 tion 2.2 and advances the previous study by (Wang & Manga, 2021). The aim is to in-268 vestigate and establish validity constraints for the use of the analytical solution when 269 interpreting the groundwater response to Earth tides. 270

When modeling Earth tides, an external gravitational strain  $\epsilon_G(x, y, z, t)$  is applied to deform the Earth's crust and the resulting fluid pressure  $p_f(x, y, z, t)$ , and displacement vector  $u_{ii}(x, y, z, t)$  is calculated. Under the free surface condition, the normal stress along the radius is zero. Hence, gravitational strain can be decomposed into its vertical  $\epsilon_h(x, y, t)$  and horizontal component  $\epsilon_v(z, t)$  (Agnew, 2005),

$$\epsilon_G(x, y, t) = \epsilon_h(x, y, t) + \epsilon_v(z, t). \tag{18}$$

Earth tides induce an *eigenstrain*, i.e., a strain that does not result directly from an applied force. Qu and Cherkaoui (2006) describes the differences and relationships between total, elastic and eigenstrains. To simulate the effect in a realistic well-aquifer system, in which the hydraulic and geomechanical properties of the material may vary in space, we applied vertical and horizontal strain as displacement boundary conditions. This fixes the internal strain throughout the model as a function of the filling material elastic tensor as (Wang, 2017),

$$r = R : u_{rr} = \epsilon_h(x, y, t) R, \qquad (19)$$

285

$$r = 0: \ u_{rr} = 0, \tag{20}$$

284 and

$$z = 0: u_{zz} = \epsilon_v(z, t) L, \qquad (21)$$

$$z = -L: \ u_{zz} = 0, \tag{22}$$

where R and L are the horizontal and vertical lengths, respectively. A constant atmospheric pressure (i.e. drained condition) is assumed at the top of the modelling domain as

$$z = 0: p_f = 0. (23)$$

The governing equations follow the traditional Biot (1941) theory of a linear elas-289 tic, saturated and deformable porous medium with water as the fluid (Wang, 2017; Cheng, 290 2016). The strong form of the general equations in Subsection 2.1 can be converted into 291 the respective weak form and discretized before solving with the finite element (FE) method. 292 In this study, we adopt the continuum representation of an elementary volume (REV) 293 in a porous medium. To solve the numerical system the Real Heterogeneity App (RHEA), 294 a FE application based on the Multiphysics Object Oriented Simulation Environment 295 (MOOSE) was used (Permann et al., 2020). A detailed description of the system of equa-296 tions to be solved as well as further information of the tight coupling and numerical de-297 scription of the FE implementation utilized in this study can be found in (Bastías Es-298 pejo et al., 2021; Wilkins et al., 2021, 2020). For consistency, we keep the original no-299 tation used in Bastías Espejo et al. (2021), where the field variables are the fluid pres-300 sure  $p_f$  and the displacement vector  $u_{ii}$ . 301

#### 302

#### 2.4 Assessing the subsurface response conditions

Within the theory of linear poroelasticity (equations 1 to 4), one can distinguish 303 between two end-members that describe the type of pore pressure response to stresses 304 and strains: undrained and drained. When a deformation in a porous medium occurs, 305 under drained conditions, the rate of applied distortion is slow in relation to the abil-306 ity of the porous medium to allow dissipation of the pressure wave. This results in the 307 flow of fluid caused by the buildup of pressure differences. Under undrained conditions, 308 the rate of applied distortion is fast enough for an instantaneous pore pressure response 309 to external deformations and fluid cannot flow in response. 310

The type of subsurface response is represented by equation 2, in which  $\xi$  describes an increment of change in fluid content. Under *undrained* conditions,  $\xi = 0$  and equation 2 reduces to equation 13. While under *drained* conditions  $\xi \neq 0$ . Hence, the confined pore pressure as a response to Earth tides can be obtained with equation 2 if the Earth tide strain ( $\epsilon_G$ ) is known. According to Rojstaczer and Agnew (1989), an areal strain is sufficient for depths that are relevant to hydrogeology.



Figure 1. Overview of the conceptual models used in this work. (a) Analytical model of Wang et al. (2018) when no external force is applied (b) Analytical model of Wang et al. (2018) when the confined pore pressure generated at the far field is generated and fluid flow towards the well. (c) A 1D column of the subsurface representative of the aquitard to assess the type of response. At t = 0 the column is equilibrium, at t > 0 a harmonic strain is applied which results in fluid movement in and out of the column. (d) A 2D model of the aquifer bounded by a aquitard and connected to a well. Earth tide strain is applied which moves fluid towards a well that is numerically modeled.

For Earth tides, the applied strain depends on the frequency of the harmonic and the type of response is a function of the hydro-geomechanical subsurface properties as well as depth. To assess the type of response under realistic conditions, we numerically modeled an infinitely long 1D (5000 m) section of the subsurface (Figure 1c) using a harmonic function as displacement boundary condition as follows

$$z = 0, \,\epsilon_{M2} = \epsilon_0 \cdot \sin\left[2\pi f_{M_2}t\right]. \tag{24}$$

Here,  $\epsilon_0$  is the amplitude of the Earth strain,  $f_{M_2}$  is the frequency of the  $M_2$  component and t is the time in days. We computed the increment of fluid content,  $\xi$ , over depth  $0 \leq z \leq 1000$  m and repeated the calculations by setting assumed, but realistic discrete values of specific storage at constant strain  $S_{\epsilon}$ ,  $(1 \cdot 10^{-7} m^{-1}, 1 \cdot 10^{-6} m^{-1} \text{ and } 1 \cdot 10^{-5} m^{-1})$ , and bulk modulus K,  $(1 \cdot 10^9 Pa, 1 \cdot 10^{10} Pa and 1 \cdot 10^{11} Pa)$  and continue values of hydraulic conductivity of the aquifer,  $1 \cdot 10^{-7} \leq k_a \leq 1 \cdot 10^{-3} ms^{-1}$  within the domain. These values represent realistic conditions as reported in the geoscience literature (Das & Das, 2008; Wang, 2017; Cheng, 2016; Lade, 2001).

When analyzing the groundwater response to Earth tides, undrained conditions have to be given for the analytical solution to be valid (Appendix A). However, the type of response to Earth tides has not been assessed before and is therefore unknown. To assess whether the subsurface response is sufficiently undrained, we can use equation 2 assuming  $\alpha = 1$ . Under undrained conditions  $\xi = 0$ , such condition can be assumed when  $\epsilon \gg \xi$ . Hence, the effect of  $\xi$  can be neglected in equation 2.

## 2.5 Numerical model of a coupled well-aquifer system

336

To investigate the limitations of the analytical model presented in Section 2.2, a 337 2D axial-symmetric cylindrical model was developed. The model accounts for poro elas-338 tic aquifer (Section 2.1) bounded by a low permeable aquitard on top and by a rigid aquiclude 339 on the bottom. Gravity and horizontal flows are allowed (Figure 1d). Boundary condi-340 tions are set as described in Section 2.5. The applied Earth strain ( $\epsilon_G$ ) was calculate the-341 oretically with PyGtide (Rau, 2018), a Python wrapper for ETERNA PREDICT 3.4. 342 We chose the city of Berlin (Germany) and a signal frequency of 2 cpd for simplicity as 343 it closely resembles  $M_2$  with a duration of 30 days. While the location is arbitrary, it does 344 not change the conclusions because the context of this study is generic. 345

The borehole-subsurface system is modeled as a 1D element outside the 2D system. To relate the porous medium and the borehole-subsurface, the boundary at r =0 is modeled as a free drainage boundary, i.e., a sink boundary where the flux is computed in function of the pressure at r = 0 ( $P_{r=0}$ ) and at the bottom of the boreholesubsurface ( $P_w$ )

$$r = 0, -101 \le z \le 100, \ \dot{m}_f = C(p_f - p_{r=0}),$$
(25)

where  $m_f$  is the mass flux, C the conductance (i.e., how efficiently fluid is transported though a boundary) between the borehole-subsurface and the model. For this study  $C = 10^{-3} m^2 s^{-1}$  which is high enough to ensure that  $p_f = p_{r=0}$  at the end of each non-linear iteration.

As a result, the mass flux through the boundary between the model and the well is computed in every non-linear iteration, which fixes the pore pressure at the boundary for the next iteration. This way, the fluid level in the well is tightly coupled to the pressure at the well boundary of the model as the fluid level in the well is computed in the same Jacobian matrix with the numerical model and is computed as,

$$\frac{dh_f}{dt} = \frac{\dot{m}_f A_c}{\rho_f A_w}.$$
(26)

Here,  $h_f$  represents the water level in the well  $(h_f = p_f g^{-1} \rho_f^{-1})$ ,  $A_c$  is the external area of a cylinder and  $A_w$  is the inner area of a cylinder. Since the model is linear elastic, typically, only two non-linear iterations are needed.

The model domain is R = 5000 m long in the r direction and L = 101 m in depth 363 in the z direction. The aquitard is 100 m thick, whereas the aquifer is 1 m thick, and 364 we assume that the well is screened throughout this unit. The 101 m long well is located 365 at the left boundary of the modeling domain and the well has  $r_w = 0.2$  m of radius (Fig-366 ure 1d). The geometry complies with previous studies (Hsieh et al., 1987; Wang et al., 367 2018) and therefore enables a comparison. The finite elements were discretized using the 368 built-in mesh generator of MOOSE and the element size increases logarithmically along 369 the r-axis away from the well. The mesh is vertically and logarithmically discretized 5 370 times across the aquifer, 20 times across the top layer and 100 times in the horizontal 371 direction. The material properties of the model are summarized in Table 1. The values 372

of Table 1 were assumed in previous studies (Wang et al., 2018) and extracted from literature (Das & Das, 2008; Wang, 2017; Cheng, 2016; Lade, 2001).

The initial pore pressure condition is set as  $p_f^0 = -\rho_f gz$  and the effective initial 375 stress as  $\sigma_{zz}^{\prime 0} = (\rho_s - \alpha \rho_f)gz$ , where  $\sigma_{zz}^{\prime 0}$  is the vertical component of the effective stress 376 at time zero. Again, we apply a harmonic displacement function with the  $M_2$  frequency 377 computed with a tidal catalog, the amplitude of the strain is  $\epsilon_0 = 1.2 \cdot 10^{-8}$ . The model 378 runs until it reaches quasi steady-state, at which point the well physics as well as tidal 379 forcing as boundary conditions are activated. This approach minimizes potential numer-380 ical overshooting produced by the free drainage boundary between the porous medium 381 and the well. We verify this numerical implementation using the analytical solution of 382 Wang et al. (2018) (Subsection 2.2) with the aquitard permeability set to zero, i.e., the 383 model represents only one layer. 384

#### **385 3 Results and discussion**

386

#### 3.1 Analytical solutions

Previous research has used the analytical solutions by Hsieh et al. (1987), H. F. Wang 387 (2000) and Wang et al. (2018) to estimate hydraulic properties for negative and posi-388 tive phase shifts between groundwater and Earth tides, respectively. We note that Hsieh 389 et al. (1987) and Wang et al. (2018) requires undrained conditions whereas H. F. Wang 390 (2000) assumes drained conditions. The validity conditions for these assumptions have 391 not been investigated for Earth tide frequencies. For drained conditions the relationship 392 between stress and strain is no longer linear, as porepressure also plays a role bearing 393 loads, i.e. see Equation 2. Furthermore, while H. F. Wang (2000) considers vertical flow 394 in a one dimensional poroelastic aquifer, it neglects the influence of an observation well. 395 As shown in Figure 2, a well generates phase shifts between the confined far distance pore-396 pressure and the water level in the well as the fluid requires time to move in and out of 397 the well. Strictly speaking, this solution was derived for surface loads, such as exerted 398 from atmospheric pressure, but not for Earth tide strains. These aspects illustrate that 399 H. F. Wang (2000) has limited use when estimating hydraulic properties from the ground-400 water response to Earth tides. 401

Wang et al. (2018) provides an extended formulation to Hsieh et al. (1987) con-402 sidering vertical aquitard leakage accounting for both negative and positive phase shifts. 403 It is useful to illustrate the solution space of Wang et al. (2018) by providing an overview 404 of amplitude ratios and phase shifts (Equations 14 and 15) as a function of realistic ranges 405 of the aquifer hydraulic conductivity  $(k_a)$  and specific storage at constant strain  $(S_{\epsilon})$  as 406 well as discrete values of leakage, see Figure 2. Note that this is based on the dominant 407 harmonic signal frequency of 2 cpd, a well and screen radius of 0.2 m, a screen length 408 of 1 m and an aquitard thickness of 100 m. The first row, Figure 2a and 2b, shows the case where there is no vertical leakage leading to negative phase shifts only. We confirm 410 the reports by Wang et al. (2018) that the analytical solution matches the previous so-411 lution by Hsieh et al. (1987) when the aquitard hydraulic conductivity is set to zero. 412

The solution space shows that vertical leakage causes positive phase shifts at relatively high aquifer hydraulic conductivity, i.e.,  $k_a > 1 \cdot 10^{-5} ms^{-1}$  in Figure 2d. This threshold is even more clear for vertical leakage larger than  $k_l > 1 \cdot 10^{-6} ms^{-1}$  where the transition to positive phase shift is almost linear. Moreover, in Figures 2d and 2f, the phase shift behavior is very similar for the lower part of the specific storage at constant strain under study ( $S_{\epsilon} < 1 \cdot 10^{-5} m^{-1}$ ). A similar case is observed in Figures 2c and 2e where the amplitude response of the analytical solution shows very similar results.

The solution space illustrated in Figure 2 shows that the solution is under-determined when subsurface parameters are estimated, because three hydraulic properties (hydraulic

nodel.	
umerical n	
f the r	
properties o	
mechanical	
and	
of the hydraulic	
Overview	
Table 1.	

Property	Acronyms	Value	Unit	Reference
	-	1 10-7 / 1 / 10-4	T.	
Aquitara nyaraune conductivity	$\kappa_l$	$z  01 \leq k_l \leq 10$	m s	(wang, zuit; Cheng, zuio)
Aquitard specific storage at constant strain	$S_{\epsilon,l}$	$1\cdot 10^{-7}, 1\cdot 10^{-6}, 1\cdot 10^{-5}$	$m^{-1}$	(Wang, 2017; Cheng, 2016)
Aquitard bulk modulus	$K_l$	10	GPa	(Das & Das, 2008)
Aquitard Poisson's ratio	рı	0.25	ı	(Das & Das, 2008)
Aquitard Biot coefficient	$a_l$	1	ı	(Wang, 2017; Cheng, 2016)
Aquifer hydraulic conductivity	$k_a$	$1 \cdot 10^{-3},  1 \cdot 10^{-4},  1 \cdot 10^{-5}$	$m \ s^{-1}$	(Wang, 2017; Cheng, 2016)
Aquifer specific storage at constant strain	$S_{\epsilon,a}$	$1\cdot 10^{-7},  1\cdot 10^{-6},  1\cdot 10^{-5}$	$m^{-1}$	(Wang, 2017; Cheng, 2016)
Aquifer bulk modulus	$K_a$	10	GPa	(Das & Das, 2008)
Aquifer Poisson's ratio	$ u_a$	0.25	ı	(Das & Das, 2008)
Aquifer Biot coefficient	$a_a$	1	ı	(Das & Das, 2008)
Skeleton density	$ ho_s$	2000	$kg m^{-3}$	(Wang, 2017; Cheng, 2016)
Bulk modulus of the water	$K_{f}$	2.2	GPa	(Wang, 2017; Cheng, 2016)
Water density	$\rho_f$	1000	$kg m^{-3}$	(Wang, 2017; Cheng, 2016)
The subscript $l$ and $a$ refer to aquitard and aqu	ifer, respectively			



Figure 2. Amplitude and phase shift response of the analytical solution presented in Wang et al. (2018) for a realistic range of hydraulic conductivity and specific storage at constant strain values. (a) and (b) are representative of zero leakage through the aquitard corresponding to Hsieh et al. (1987). (c) to (f) consider distinct and increasing aquitard hydraulic conductivity values. The harmonic signal frequency is 2 cpd, the well and screen radius are 0.2 m, the screen length is 1 m and the aquitard has 100 m thickness

- conductivity of the aquifer  $k_a$ , vertical leakage of the aquitard  $k_l$  and specific storage at 423 constant strain  $S_{\epsilon}$ ) are derived from only two equations (equations 14 and 15) causing 424 non-unique solution pairs. Further, solving for the parameters is an optimization prob-425 lem that requires root finding through iteration. The search is based on an initial guess 426 and the method may find different potential solutions depending on the initial guess. We 427 investigate the effect of the initial guess on the solution space by visualizing the non-uniqueness. 428 Figure 3 shows the variability in estimated parameters due to providing different initial 429 guesses for least-squares solving. Blue color means less variability whereas red color shows 430 a larger difference in the solution space and therefore stronger non-uniqueness. 431
- In general, higher sensitivity to the initial values is observed in the phase shifts compared to the amplitude ratios. The sensitivity for inverting the hydraulic conductivity of the aquifer  $k_a$  is relatively low ( $\Delta k_a < 5 \cdot 10^{-4} m s^{-1}$ ) at negative phases and in-

creases as the phase shift approaches 0°. For values higher than zero degrees the sensitivity decreases again until approximately 40° where it starts to increase again (Figure 3a). The highest sensitivity is found around amplitude ratio of one and positive phase
shift. Situations where the amplitude ratio is one and the phase shift much higher than
zero are not realistic and should be disregarded.

Specific storage at constant strain shows a high contrast in solution variability with 440 values that are very close to  $\Delta S_{\epsilon} = 1 \cdot 10^{-5} m^{-1}$  for most of the solution space (Fig-441 ure 3b). At low phase shifts ( $\Delta \phi < -70^{\circ}$ ) the variability significantly reduces to  $\Delta S_{\epsilon} =$ 442  $1 \cdot 10^{-7} m^{-1}$ . In practice, most of the realistic cases will fall within the high variabil-443 ity zone. Vertical leakage shows relatively low sensitivity where the phase shift is neg-444 ative ( $\Delta k_l < 5 \cdot 10^{-5} \text{ ms}^{-1}$ , Figure 3c). However, at positive phase sifts the variabil-445 ity increases up to two orders of magnitude, demonstrating the effect of the phase shift 446 on vertical leakage. 447

In the illustrated case, the sensitivity of the specific storage at constant strain is 448 constant throughout the solution space and therefore its initial value does not play a sig-449 nificant role on finding different solutions. Negative phase shifts show a low sensitivity 450 to the initial condition and will likely result in an accurate inversion of the hydraulic prop-451 erties without a priori knowledge of the subsurface properties as mentioned in section 452 2.2. For positive phase shifts a handle on at least one of the properties is necessary as 453 the vertical leakage significantly increases its variability. Further, the hydraulic conduc-454 tivity increases its variability towards high amplitudes which is the range where Earth 455 tide methods work best. Overall, this complies with the previous finding that positive 456 phase shifts can robustly be interpreted as vertical leakage (Wang et al., 2018). 457



Figure 3. Color map exploring the solution space, i.e., the variability of parameters as a function of the initial guess, of the under-determined problem by (Wang et al., 2018). Estimating three hydraulic properties out of two measured parameters: (a) aquifer hydraulic conductivity, (b) specific storage at constant strain, (c) aquitard hydraulic conductivity. Note that each color scale has a different range. Blue indicates less variability, whereas red means more variability of the results.

#### 3.2 Notes on the specific storage

458

An interesting implication of Section 2.1 arises when  $\alpha = 1$ . The latter refers to systems where the compressibility of grains is small compared to the compressibility of grains, such as is the case for unconsolidated materials. Here, the specific storage at constant strain (the inverse of the *Biot* modulus, equation 3) reduces to

$$S_{\epsilon} = \frac{1}{M} = \frac{n}{K_f}.$$
(27)

Since the bulk modulus of water is known ( $K_f = 2.2 \cdot 10^9 Pa$ ), the porosity of the material can also be estimated from the groundwater response to Earth tides. However, for consolidated materials the Biot coefficient may be smaller than one. This can help to constrain the expected values of the specific storage at constant strain. For instance, if the subsurface material is unconsolidated and has realistic porosity values, i.e.,  $0.01 \leq$  $n \leq 0.3$ , then the specific storage at constrained to

$$4.5 \cdot 10^{-8} m^{-1} \le S_{\epsilon} \le 1 \cdot 10^{-6} m^{-1}.$$
<sup>(28)</sup>

We note that previous studies which estimated the specific storage from the ground-469 water response to Earth tides have not considered the appropriate context for this prop-470 erty. The result is referred to as "specific storage at constant strain"  $(S_{\epsilon})$  and it can vary 471 significantly from the specific storage generally used in hydrogeology  $(S_s, \text{ see Equation})$ 472 9) (Hantush, 1960). The difference between both coefficients originates from the under-473 lying assumptions. The specific storage at constant strain is defined in conditions in which 474 the volume of the porous frame is maintained constant but the fluid volume is not, which 475 induces changes in the pore volume because fluid has to be accommodated. In contrast, 476 for the specific storage used in hydrogeology the porous frame is allowed to deform in 477 the vertical direction. This is mathematically represented by the second term of Equa-478 tion 9. Thus, when the subsurface material is much less compressible than the filling fluid 479 and the pore space the second term of Equation 9 tends to zero because no deformation 480 of the frame takes occurs, hence  $S \approx S_{\epsilon}$ . As demonstrated in this study, attention must 481 be paid to the conceptual difference between these two parameters. 482

483

#### **3.3** Numerical modeling of the groundwater response to Earth tides

The fluid continuity equation (equation 4) has been solved in previous studies as-484 suming that the strain term  $\epsilon(t)$  is known and solely time-dependent with adequate bound-485 ary conditions. This equation is an *inhomogenous* diffusion equation for which the change 486 of volumetric strain is mathematically equivalent to a sink/source in the aquifer stor-487 age term. Therefore, changes of strain result in changes of porepressure in the entire model 488 domain. When mechanical coupling is included, the continuity equation needs to be cou-489 pled to the state of stress, hence the strain is tightly coupled to porepressure. Thus, the 490 strain term in equation 4 is no longer uniform over the entire model and may vary de-491 pending on the amount of change of fluid. For instance, changes in porepressure (for earth 492 tides within the radius of influence of the well) induces changes in the volumetric strain, 493 which generate drained conditions (Section 3.4). Therefore, assuming that the applied 494 strain is constant within the model domain is inaccurate for our purposes since, as ex-495 plained before, the strain is function of the porepressure. Another common way to ex-496 press the coupling between porepressure and strain is by rearranging Equations 1 and 497 2 as498

$$\epsilon = \frac{1}{K}\sigma + \frac{\alpha}{K}p_f.$$
(29)

The relative movement of celestial bodies in relation to Earth induces variations 499 in the gravitational force which results in small deformation of the Earth's crust. Such 500 deformations are not caused by an applied stress. In continuum mechanics, this prob-501 lem is known as *eigenstrain*, and it is very common in heat transport, e.g., dilatation caused 502 by heating of materials. The relationship between deformation and *eigenstrain* can be 503 obtained experimentally leading to a constitutive model (Qu & Cherkaoui, 2006). In this 504 work we apply a simpler approach. As FE implementation typically requires displace-505 ment or load as boundary conditions, we set displacement as boundary condition and 506 directly applied the strain obtained from an Earth tide catalog multiplied with the length 507 of the model, i.e.,  $u_{ii} = \epsilon_{ii}L$ . 508

While this approach is convenient it has limitations. If the mechanical properties change over the modeling domain (composite material), the displacement will not be uniformly distributed across the domain and therefore the resulting strain will also be nonuniform. This would produce larger displacements in soft layers resulting in higher porepressure. One way to solve this problem is to assume vertical heterogeneity and to apply the total volumetric strain only at the horizontal boundaries. This would result in a uniform displacement distribution in the horizontal axis and therefore result in an appropriate porepressure response. We note that the effects of distributed mechanical heterogeneity are not further explored in this work.

Initialization of the numerical model is not trivial since the initial hydrostatic and 518 mechanical states (initial porepressure and stresses) has to be in equilibrium (Settari & 519 Walters, 2001). This challenge applies in particular for heterogeneous distributions of 520 material properties and transient boundary conditions. Achieving mechanical equilib-521 rium at time t = 0 is difficult and may in most cases require a separate initialization 522 step during the simulation (Chen et al., 2009). We recommend to first simulate steady-523 state conditions which generates the stress and porepressure distribution within the mod-524 eling domain. 525

From a numerical point of view, the simulator is setting the force balance as an ap-526 proximation, i.e.,  $\nabla \sigma = 0$ . In practice, a non-linear step is finished when the force bal-527 ance falls below a threshold close to zero but residual errors always remain. Earth tides 528 generate only small changes in porepressure which are close to the residual error. For 529 example, if the acceptable error is  $e = 1 Pam^{-2}$ , then in our case the area is 5050000 530  $m^2$  leading to total residuals up to  $R \approx 5050 k P a$  at the bottom of the model. Since 531 Earth tides generate porepressure change in smaller magnitude, minimizing the error is 532 an important consideration when modeling. Numerical modeling of Earth tides there-533 fore requires attention to decreasing the tolerance of the numerical solver (e.g., by in-534 creasing the number of linear steps), increasing space discretization (e.g., by increasing 535 the size of the Jacobian matrix) or decreasing time discretization (e.g., by increasing the 536 number of time steps). 537

#### 538

#### 3.4 Are conditions for the $M_2$ Earth tide drained or undrained?

When a stress is applied to an undrained subsurface system, the load is shared by 539 the bulk material, the grains and the pore fluid. The balance between these three responses 540 results in instantaneous deformation of the pore space and a change in fluid pressure. 541 If the rate of the applied deformation is slow enough then fluid can flow out of the sys-542 tem which reduces the porepressure. The balance between the rate of Earth tide stress 543 and realistic hydro-geomechanical subsurface properties is rarely known. Moreover, fluid 544 movement (i.e., drained conditions) may be given leading to  $\xi \neq 0$ . Under such con-545 ditions the assumptions of the analytical solutions are violated potentially leading to er-546 rors when interpreting the groundwater level response to Earth tides. 547

To assess the conditions under which an undrained response occurs, we numerically simulate a 1D vertical column with depth  $0 \le z \le 5000$  m (Figure 1b) and with a range of realistic hydraulic and geomechanical properties. Equation (2) can be solved for fluid quantity ( $\xi$ ) assuming the worst scenario ( $\epsilon_G = 1 \cdot 10^{-8}$  corresponding to a low tide amplitude) and  $\alpha = 1$ ,

$$\xi = \frac{p_f}{M} - \epsilon_G = S_\epsilon p_f - \epsilon_G,\tag{30}$$

Figure 4 shows the results of our numerical model which calculates  $\xi$  up to 1000 m depth and for a range of realistic hydraulic properties as well as discrete values of the bulk modulus. As typical Earth tide amplitudes vary between  $1 \cdot 10^{-7} \le \epsilon_G \le 1 \cdot 10^{-8}$  (Rojstaczer & Agnew, 1989), we define

$$\mathcal{E} < 5 \cdot 10^{-11} \tag{31}$$

as a condition for an undrained response for which the analytical solution is valid, i.e., no porepressure changes occur under this value. This is highlighted in Figure 4 and allows an assessment of the conditions for which the analytical solution should be valid.



Figure 4. Change of fluid content  $\xi$  over depth and aquitard hydraulic conductivity for a 1D column (Figure 1b) and Earth tide forcing with  $M_2$  frequency. Rows correspond to different values of specific storage whereas columns are representative for different bulk moduli, e.g., clay (a, d, g), sand (b, e, h) and hard rock (c, f, i). Values of  $\xi$  can be used to infer the depth at which the system response is undrained, i.e., where application of the analytical solution (Equations 14 and 15) is valid. Value ranges of validity are delineated by the dashed line.

Figure 4 shows that undrained conditions are more likely the deeper a system. Fur-560 ther, when the hydraulic conductivity of the leaky layer  $(k_l)$  increases, the system be-561 haves more drained. This is expected as the system becomes more permeable and there-562 fore allows flow in response to pressure gradients. This results in fluid movement which 563 causes increased drainage. Similarly, as the specific storage at constant strain increases 564 (rows of plots in Figure 4) the level of drainage decreases. This is because as  $S_{\epsilon}$  increases 565 the volume of fluid that the system contains due to deformation increases leading to less 566 fluid moving out of the system. This can also be explained using Equation 4 when di-567 viding by  $S_{\epsilon}$ 568

$$\left(\frac{\partial p}{\partial t} + BK_u \frac{\partial \epsilon}{\partial t}\right) = \frac{k}{S_\epsilon} \nabla^2 p + \frac{Q}{S_\epsilon}.$$
(32)

Equation 32 illustrates that the hydraulic diffusivity of a system  $(kS_{\epsilon}^{-1})$  decreases with the increase of  $S_{\epsilon}$ .

As the bulk modulus (K) increases, see columns in Figure 4, the system becomes more drained. An explanation for this is that as the filling material becomes stiffer, the mechanical coupling becomes less relevant and the system approaches an incompressible porous skeleton. Under such conditions, only a drained response is allowed and an instantaneous pneumatic response of the system is no longer possible. This can also be explained by revisiting the definition of the Skempton coefficient (B). Assuming  $\alpha =$ 1, then

$$B = \frac{K_f}{K_f + \phi K} \tag{33}$$

which illustrates that when the bulk modulus increases *B* decreases. This results in a reduction of the overall storativity of the system and consequently also drainage. Another way to understand this result is by considering the coupling of equations. For this simulation we assumed mechanical stress balance as follows

$$\frac{\partial \sigma_{zz}}{\partial z} = 0. \tag{34}$$

Here, Equation 1 must remain constant when the total stress increases (first two terms of the Equation 1) thus the amount of fluid leaving the system must increase (third term of the equation 1).

In general, a larger porosity will increase the value of the specific storage at con-585 stant strain, which will decrease the level of drainage. Our assessment shows that when 586 the hydraulic conductivity of the leaky layer exceeds  $10^{-5} ms^{-1}$ , this leads to drained 587 conditions and could result in errors when the analytical solution is used to estimate the 588 properties of the aquifer. However, it is worth noting that the level of drainage depends 589 on the geomechanical properties of the system, as well as depth and frequency of the sig-590 nal. The amplitude of the signal,  $\epsilon_0$ , for field measurements, as higher amplitudes will 591 generate higher confined porepressure and facilitate detection of fluid level changes in-592 side the observation well. 593

As shown in Figure 4, the level of confinement depends on the hydraulic and geomechanical properties of the subsurface under consideration. Consequently, defining conditions under which an undrained response exists depends on the particular field conditions, e.g., depth of the borehole and some knowledge of the subsurface properties. Figure 4 can be used as a preliminary guide for assessing whether or not it is appropriate to apply the analytical solution for interpreting the groundwater level response to Earth tides.

601

## 3.5 Validity and limits of analytical Earth tide interpretations

Determining subsurface properties from Earth tide responses requires system confinement as a basic condition. To study the effects of a realistic well-aquifer system and the effect of (un)drained conditions, the level of confinement is gradually relaxed in a layered 2D model in this section. Figure 5 shows the results from our numerical model (Section 2.5) compared to the analytical solution without vertical leakage ( $k_l = 0 m s^{-1}$ ) for a tidal signal with 2 cpd frequency (Section 2.2). The good agreement of amplitude ratios and phase differences verifies our coupled numerical modeling approach. This allows a rigorous hydraulic and geomechanical assessment of how realistic conditions (e.g., subsurface layered heterogeneity) affects the groundwater response to Earth tides.



**Figure 5.** Verification of the 2D numerical model against Hsieh et al. (1987) for a harmonic forcing signal with 2 cpd frequency. Here, the simulation of the hydraulic conductivity of the leaky layer was set to zero.

611	Numerical simulations consider discrete values of the hydraulic conductivity of the
612	aquifer $(10^{-3} ms^{-1}, 10^{-4} ms^{-1} \text{ and } 10^{-5} ms^{-1})$ and varying values of the aquitard, $10^{-7} \leq 10^{-7}$
613	$k_l \leq 10^{-4}$ (in $ms^{-1}$ ). In addition, discrete values of specific storage at constant strain
614	$(10^{-5} m^{-1}, 10^{-6} m^{-1} \text{ and } 10^{-7} m^{-1})$ were investigated. For detailed information of all
615	the material parameters used in the simulation please refer to Table 1.

The effect of the amplitude ratio and the phase shift due to leakage of the aquitard are shown in figure 6. The columns represent values of aquifer hydraulic conductivity  $(k_a)$ . The first row shows the effect on the amplitude ratio (A) and the second column the effect of the phase shift ( $\phi$ ) over different levels of aquitard confinement  $(k_l)$ . Each line in the figure 6 correspond to the three discrete simulated values of the specific storage at constant strain ( $S_{\epsilon}$ ). Simulations are shown with marked lines while the analytical solution of Wang et al. (2018) is shown with dashed lines.

The level of drained conditions can be assessed in conjunction with figure 4 for the three specific storage at constant strain simulated here (figure 4b, 4e and 4h). For example, at 100 m depth (which is the thickness of the aquitard in the simulations) and  $S_{\epsilon} = 10^{-7} m^{-1}$  (Figure 4b), the system shows drained conditions within the entire range of confinement  $(10^{-7} \le k_l \le 10^{-4} ms^{-1})$ , see the blue triangle markers in Figure 6). Therefore, the simulated amplitude under these conditions is somewhat lower and the



Figure 6. Comparison of the amplitude ratios and phase shifts obtained from numerical modeling and the analytical solution by Wang et al. (2018).

phase shift higher compared to the analytical solution. This results in an underestimation of the hydraulic properties of the aquifer  $(k_a \text{ and } S_{\epsilon})$  or overestimation of leakage from the aquitard  $(k_l)$  when the analytical solution is used.

<sup>632</sup> When the specific storage at constant strain of the aquifer is  $S_{\epsilon} = 10^{-6} m^{-1}$ , at <sup>633</sup> 100 m depth, the system is in a transition zone between positive and negative change <sup>634</sup> of fluid content when  $k_l = 10^{-7} ms^{-1}$  (Figure 4e). Since the simulated amplitude ra-<sup>635</sup> tio and phase shift match the analytical solution, the system can still be assumed as undrained <sup>636</sup> within this transition zone. However, for higher levels of leakage ( $k_l > 10^{-7} ms^{-1}$ ), the <sup>637</sup> system is outside this transition zone and completely drained conditions prevail. The lat-<sup>638</sup> ter leads to significant differences between numerical and analytical results.

Similar results are observed when the specific storage at constant strain of the aquifer is  $S_{\epsilon} = 10^{-5} m^{-1}$ . When the level of leakage is  $k_l \leq 5 \cdot 10^{-5} ms^{-1}$  the system is undrained or in the transition zone and the numerical results comply with the analytical solution. For lower levels of confinement (i.e.  $k_l > 5 \cdot 10^{-5} ms^{-1}$ ), the system becomes drained and the simulation results, once again, differ compared to the analytical solution.

In the particular case when the hydraulic conductivity of the aquifer is  $k_a = 10^{-5}$  $ms^{-1}$ , the numerical result do not comply with the ones obtained with the analytical solution even under undrained conditions (Figure 6c and 6f). As the hydraulic conductivity of the aquifer decreases, the finite time to move fluid in or out of the well increases. Hence, it is likely that the fluid velocity is much more influenced by high gradients generated by the drained top boundary rather than by the gradients produce inside the open well under such conditions.

In all three columns of Figure 4, as the level of confinement provided by the aquitard decreases, the simulated results of phase shift tend towards the same value (90° for the

simulated system). This means that, as the drainage from the aquitard increases, the 653 effect of the top drained boundary over the porepressure in the aquifer increases. The 654 same effect is observable on the amplitude ratio, where the final value of the amplitude 655 ratio is the same in every specific storage at constant strain under study. This effect in-656 dicates that the hydraulic conductivity of the aquifer loses relevance as the drainage from 657 the aquitard increases. And, therefore, if the system is draining, at low levels of confine-658 ments  $(k_l > 5 \cdot 10^{-5} m s^{-1})$ , the groundwater level measured in the field can poten-659 tially result in very similar values regardless of the aquifer hydraulic conductivity. 660

661 The effect of drained conditions on the amplitude ratio and phase shift can be better understood when streamlines (i.e., the Darcy velocity field) of the system are plot-662 ted. Figure 7 shows streamlines in an area close to the open well when the amplitude 663 of the Earth tide strain is at maximum. This provides understanding of how flow paths 664 change as the level of confinement decreases at fixed aquifer specific storage at constant 665 strain  $(S_{\epsilon} = 10^{-6} m^{-1})$  and hydraulic conductivity  $(k_a = 10^{-4} m s^{-1})$ . At the high 666 confinement  $(k_l = 10^{-7} ms^{-1})$  the flow within the aquifer is horizontal and porepres-667 sure gradients are directed towards the well. This complies with the assumption of hor-668 izontal flow inherent to the analytical solution. As confinement decreases (i.e., increas-669 ing leakage of the aquitard), the velocity field shows increasing flow in the vertical di-670 rection through layers which reduces the radius of influence of the well. With the small-671 est confinement investigated (Figure 6c), vertical flow dominates in the aquifer and al-672 most no horizontal flow is observable. This shows that the pressure wave produced by 673 the open top boundary has strong effects on the amplitude ratio and phase shift at low 674 confinement and dampens the porepressure signal generated by Earth tides. 675



**Figure 7.** Streamlines show the velocity field towards the observation well during maximum Earth tide strain for three different aquitards with varying hydraulic conductivities: (a) low, (b) medium, (c) high. This illustrates that the flow direction changes from horizontal to vertical as the leakage increases.

Figure 2 can be used in conjunction with the results shown in Figure 6 to assess the potential error due to requiring undrained conditions when the analytical solution is utilized. For example, assuming a typical specific storage at constant strain of  $S_{\epsilon} = 10^{-6} m^{-1}$ , leakage of  $k_l = 10^{-6} ms^{-1}$  and hydraulic conductivity of the aquifer  $k_a =$  <sup>680</sup>  $10^{-3} ms^{-1}$ , the simulated amplitude ratio is 0.61 and the phase shift is 55.0 ° (Figure <sup>681</sup> 6a). Figures 2e and 2f show that for that amplitude ratio and phase shift, the hydraulic <sup>682</sup> conductivity of the aquifer is between  $k_a = 3 \cdot 10^{-4} ms^{-1} \le k_a \le 5 \cdot 10^{-2} ms^{-1}$ , while <sup>683</sup> the specific storage at constant strain between  $7 \cdot 10^{-5} m^{-1} \le S_{\epsilon} \le 10^{-6} m^{-1}$ . This <sup>684</sup> range is comparable with traditional field methods that estimate subsurface properties, <sup>685</sup> for example applying hydraulic tests.

Our results show that a high specific storage at constant strain  $(S_{\epsilon} \ge 10^{-6} m^{-1})$ 686 in combination with a high confinement  $(k_l \leq 5 \cdot 10^{-5} m s^{-1})$  and hydraulic conduc-687 tivity of the aquifer  $(k_a \ge 10^{-4} m s^{-1})$  allow application of the analytical solution. Ap-688 plication to real world system further requires a high contrast in hydraulic conductiv-689 ity between the layers  $(k_a k_l^{-1} \ge 10^3)$  with specific storage values that are typical ( $\approx 10^{-6}$ 690  $m^{-1}$ ). In reality, the confined porepressure is damped by the movement of fluid and fully 691 undrained conditions may be rare. Any a priori knowledge of the formation (e.g., thick-692 nesses and hydraulic properties) is key in the assessment of Earth tidal analysis, not only 693 to have a good approximation when inverting equations, but also to approximate the level 694 of drainage and therefore assess potential errors when the analytical solutions are utilized. 696

For unconsolidated systems the soil matrix is more compressible than the grains 697 (i.e., an unconsolidated subsurface) which leads to  $\alpha = 1$ . Moreover, hydraulic conduc-698 tivity and porosity for what can be considered an aquifer varies between  $10^{-2} m s^{-1} \leq$ 699  $k_a \leq 10^{-4} ms^{-1}$  and  $0.2 \leq n \leq 0.3$ , respectively. This means that the specific stor-age at constant strain varies between  $9.1 \cdot 10^{-7} m^{-1} \leq S_{\epsilon} \leq 1.4 \cdot 10^{-6} m^{-1}$  (with the bulk modulus of water,  $K_f = 2.2 \cdot 10^{-9} Pa$ ). Considering these ranges and given suf-700 701 702 ficiently high confinement between layers, application of the analytical solution to well 703 fluid levels is valid. For example, this would be the case for hydraulic properties of sands 704 and gravels overlain by clays or silts. 705

For consolidated systems, Earth tidal analysis poses a challenge as the Biot coef-706 ficient generally is  $\alpha < 1$ . In order to use the groundwater response to Earth tides, the 707 Biot coefficient has to be known as it directly attenuates the porepressure response to 708 strain (Equation 2). Although some values of the Biot coefficient have been reported for 709 different rock types varying from 0.1 to 1 (Cheng, 2016), real world measurements are 710 difficult to find in the literature (C. Wang & Zeng, 2011; Cosenza et al., 2002). Our work 711 shows that the Biot coefficient requires estimation when the groundwater response to 712 Earth tides is quantitatively evaluated for wells screened in consolidated systems. Hence, 713 for real systems, this leads to the following trade-off: As deeper wells are more likely to 714 contain Earth tide influences because undrained conditions exists, but they are also more 715 likely consolidated, in which case an estimate of the Biot modulus is required. Overall, 716 our results show that a presence of Earth tide components in wells that are screened in 717 deep and unconsolidated systems are likely to have undrained conditions and are there-718 fore suitable for interpretation. 719

#### 720 4 Conclusions

The amplitude and phase of the groundwater response to harmonic Earth tide com-721 ponents can be used to estimate hydraulic conductivity and specific storage values of aquifer 722 systems. However, this approach is based on simplified analytical solutions to the ground-723 water flow equation, which has various assumptions that have not been tested yet. To 724 assess the effect of such assumptions, we present a numerical method to simulate the ground-725 water response to Earth tides by coupling compressible flow to geomechanics. We demon-726 strated that this can be solved numerically using the Multi Object Oriented Simulation 727 Environment (MOOSE) and verify this using a simplified analytical solution of the ground-728 water flow equation. We further use simulations to assess the conditions of validity for 729

# simplified analytical solutions when estimating hydraulic properties from the groundwa ter response to Earth tides.

By first focusing on the aquitard layer, we assess the subsurface response type, i.e., 732 drained or undrained conditions, to the dominant harmonic Earth tide component at  $M_2$ 733 with frequency of 1.93227 cycles per day (cpd) for depths up to 5 km and a range of hy-734 draulic conductivities. Based on typical Earth tide strains, we define that undrained con-735 ditions exist when the incremental of fluid content is smaller than  $5 \cdot 10^{-11}$  for which 736 the groundwater equation and associated analytical solution should be valid. Our results 737 show that this is the case for specific storage at constant strain larger than  $1 \cdot 10^{-6} m^{-1}$ 738 and depths higher than 50 *m* for low conductivity systems  $(k_a < 10^{-7} m s^{-1})$  and depths up to 1 km for high conductivity systems  $(k_a \ge 10^{-3} m s^{-1})$ . 739 740

We revisited previously interpretations based on analytical solutions and showed 741 that the specific storage has been often misinterpreted. Moreover, non-uniqueness could 742 influence estimation of values of the three hydraulic properties such as (1) aquifer hy-743 draulic conductivity, (2) specific storage at constant strain and (3) aquitard hydraulic 744 conductivity. A comparison between the analytical solution and a 2D two-layered aquitard-745 aquifer system coupled to a well shows that amplitudes and phases diverge when the hy-746 draulic conductivity contrast between aquifer and aquitard reduces. This is caused by 747 decreasing confinement leading to flow paths that change from horizontal to vertical as 748 the vertical leakage increases. Applicability of the analytical solution to real-world prob-749 lems requires a hydraulic conductivity contrast of at least 3 orders of magnitude. 750

Overall, the confined porepressure generated by Earth tide strains can be signif-751 icantly attenuated by the movement of fluid through boundaries (i.e., drained conditions). 752 Furthermore, any additional a priori knowledge about the hydraulic or geomechanical 753 properties of the subsurface formation is crucial, if the groundwater response to Earth 754 tides is evaluated using analytical solutions. Our numerical approach developed and doc-755 umented can be extended to investigate the influence of other variables on results from 756 analytical solutions. Finally, results obtained from the groundwater response to Earth 757 tides should be validated with established hydraulic and geophysical methods. 758

## <sup>759</sup> Appendix A Response of well water levels to harmonic forcing

Hsieh et al. (1987) assumed unidirectional radial flow to a well which changes the
water level in a well located at a boundary of the aquifer. The head gradient in the aquifer
is given by the volumetric strain of an Earth tide which is assumed to be known. Later
on, Wang et al. (2018) complemented Hsieh et al. (1987) work by considering a two layered system by adding a leaking term to the equation 12 expressed by

$$Q = -\frac{k_l}{H_l}h,\tag{A1}$$

<sup>765</sup> with boundary conditions given by

$$t > 0, r = r_{\infty} : h(r, t) = h_{\infty} \tag{A2}$$

766

$$t > 0, r = r_w : h(r, t) = h_w(t)$$
 (A3)

$$t > 0, r = r_w : 2\pi r_w T(\partial h/\partial r) = \pi r_c^2(\partial h_w/\partial t)$$
(A4)

where  $k_l$  is the hydraulic conductivity of the leaky layer and  $H_l$  the thickness of the leaky layer. Wang et al. (2018) presented a solution for changes in well water levels are given

770 by

$$h_w = \frac{i\omega S_t}{(i\omega S_t + k_l/H_l)\gamma} \left(\frac{BK_u\epsilon_0}{\rho g}\right),\tag{A5}$$

where  $\omega$  is the angular frequency,  $h_{w,e}$  is the change in water level in the well caused by 771 Earth tides,  $\epsilon_0$  the amplitude of the Earth strain 772

$$\gamma = 1 + \left(\frac{r_c}{r_w}\right)^2 \frac{i\omega r_w}{2T\beta} \frac{K_0(\beta r_w)}{K_1(\beta r_w)},\tag{A6}$$

where  $r_w$  is the well radius,  $r_c$  is the radius of the well case,  $K_0$  and  $K_1$  are the mod-773 ified Bessel functions of the first and second kind respectively and 774

$$\beta = \left(\frac{k_l}{TH_l} + \frac{i\omega S_t}{T}\right)^{0.5}.$$
(A7)

775

799

800

801

#### Acknowledgments 776

This project has received funding from the German Research Council (DFG) grant agree-777 ment number 424795466. The authors wish to thank Dr. Andrew Wilkins for the tech-778

nical support in the numerical implementation of this study. 779

#### References 780

- Abarca, E., Karam, H., Hemond, H. F., & Harvey, C. F. (2013). Transient ground-781 water dynamics in a coastal aquifer: The effects of tides, the lunar cycle, and 782 the beach profile. Water Resources Research, 49(5), 2473–2488. 783
- Agnew, D. C. (2005). Earth tides: an introduction. University of California, San 784 Diego. 785
- Alcaraz, M., Carrera, J., Cuello, J., Guarracino, L., & Vives, L. (2021). Determining 786 hydraulic connectivity of the coastal aquifer system of la plata river estuary 787 (argentina) to the ocean by analysis of aquifer response to low-frequency tidal 788 components. Hydrogeology Journal, 29(4), 1587–1599. 789
- Allègre, V., Brodsky, E. E., Xue, L., Nale, S. M., Parker, B. L., & Cherry, J. A. 790 Using earth-tide induced water pressure changes to measure in situ (2016).791 permeability: A comparison with long-term pumping tests. Water Resources 792 Research, 52(4), 3113–3126. 793
- Bastías Espejo, J. M., Wilkins, A., Rau, G., & Blum, P. (2021).Rhea v1. 0: En-794 abling fully coupled simulations with hydro-geomechanical heterogeneity. Geo-795 scientific Model Development Discussions, 1–21. 796
- Biot, M. A. (1941). General theory of three-dimensional consolidation. Journal of 797 applied physics, 12(2), 155-164. 798
  - Chen, Y., Zhou, C., & Jing, L. (2009). Modeling coupled thm processes of geological porous media with multiphase flow: theory and validation against laboratory and field scale experiments. Computers and Geotechnics, 36(8), 1308–1329.
- Cheng, A. H.-D. (2016). Poroelasticity (Vol. 27). Springer. 802
- Cooper Jr, H. H., Bredehoeft, J. D., & Papadopulos, I. S. (1967).Response of a 803 finite-diameter well to an instantaneous charge of water. Water Resources Re-804 search, 3(1), 263–269. 805
- Cosenza, P., Ghoreychi, M., De Marsily, G., Vasseur, G., & Violette, S. (2002). The-806 oretical prediction of poroelastic properties of argillaceous rocks from in situ 807 specific storage coefficient. Water Resources Research, 38(10), 25–1. 808
- Cutillo, P. A., & Bredehoeft, J. D. (2011, 7). Estimating Aquifer Properties from 809 the Water Level Response to Earth Tides. Ground Water, 49(4), 600–610. 810 Retrieved from http://doi.wiley.com/10.1111/j.1745-6584.2010.00778.x 811 doi: 10.1111/j.1745-6584.2010.00778.x 812
- Das, B. M., & Das, B. (2008). Advanced soil mechanics (Vol. 270). Taylor & Francis 813 New York. 814
- Flemisch, B., Darcis, M., Erbertseder, K., Faigle, B., Lauser, A., Mosthaf, K., ... 815 816
  - others (2011). Dumux: Dune for multi-{phase, component, scale, physics,...}

817	flow and transport in porous media. Advances in Water Resources, $34(9)$ , $1102-1112$
818	Freeze R $\Delta$ & Cherry I $\Delta$ (1979) Crowndwater
819	$C_{200}$ X Sate K fr Horne R N (2020) Constal solution for tidal behavior in con-
820	fined and somiconfined acuifors considering skin and wallbore storage effects
821	Water Resources Research 56(6) e2020WB027195
822	Cuo H. Brodsky F. Cookel T. & Cladenhos T. (2021). Measuring fault zone and
823	host rock hydraulic proportion using tidal rosponsos Coonhusical Research Let
824	$t_{org} = \frac{18}{13} - 2021 \text{CL} 003086$
825	Haptush M S (1960) Modification of the theory of leaky aquifers Lowroad of $Geo_{-}$
820	nhusical Research 65(11) 3713–3725
021	Herrera-García G. Ezquerro P. Tomás B. Béjar-Pizarro M. López-Vinielles I.
820	Rossi M others (2021) Mapping the global threat of land subsidence
830	Science, 371(6524), 34–36.
831	Hsieh P A Bredehoeft J D & Farr J M (1987) Determination of acquifer
832	transmissivity from earth tide analysis. Water resources research, 23(10).
833	1824–1832.
834	Jacob, C., & Lohman, S. (1952). Nonsteady flow to a well of constant drawdown
835	in an extensive aquifer. Eos. Transactions American Geophysical Union, 33(4).
836	559–569.
837	Jacob, C. E. (1946). Radial flow in a leaky artesian aquifer. <i>Eos. Transactions</i>
838	American Geophysical Union, 27(2), 198–208.
839	Jardani, A., Dupont, JP., Revil, A., Massei, N., Fournier, M., & Laignel, B. (2012).
840	Geostatistical inverse modeling of the transmissivity field of a heterogeneous
841	alluvial aquifer under tidal influence. Journal of Hydrology, 472, 287–300.
842	Keilegavlen, E., Berge, R., Fumagalli, A., Starnoni, M., Stefansson, I., Varela, J.,
843	& Berre, I. (2021). Porepy: An open-source software for simulation of mul-
844	tiphysics processes in fractured porous media. Computational Geosciences,
845	25(1), 243-265.
846	Kolditz, O., Bauer, S., Bilke, L., Böttcher, N., Delfs, JO., Fischer, T., others
847	(2012). Opengeosys: an open-source initiative for numerical simulation of
848	thermo-hydro-mechanical/chemical $(thm/c)$ processes in porous media. Envi-
849	ronmental Earth Sciences, $67(2)$ , 589–599.
850	Lade, P. V. (2001). Engineering properties of soils and typical correlations.
851	In Geotechnical and geoenvironmental engineering handbook (pp. 43–67).
852	Springer.
853	Lambe, T. W., & Whitman, R. V. (1991). Soil mechanics (Vol. 10). John Wiley &
854	Sons.
855	McMillan, T. C., Rau, G. C., Timms, W. A., & Andersen, M. S. (2019, 6). Uti-
856	lizing the Impact of Earth and Atmospheric Tides on Groundwater Systems:
857	A Review Reveals the Future Potential. Reviews of Geophysics, $57(2)$ , 281–
858	315. Retrieved from https://onlinelibrary.wiley.com/doi/abs/10.1029/
859	2018 RG000630 doi: $10.1029/2018 RG000630$
860	Meinzer, O. E. (1939). Ground water in the United States, a summary of ground-
861	water conditions and resources, utilization of water from wells and springs, methods of scientific investigation, and literature relating to the subject (
862	ad: Tach Ban) US C PO Batriovad from http://pubs.or.uggg.gov/
863	publication/wsp836D
804	Morritt M I (2004) Estimating hydraulic properties of the $\{Floridan\}$ against
866	sustem by analysis of earth-tide ocean-tide and harometric effects [ Collier
867	and Hendry Counties, Florida (Tech Rep No 034267) US Geol Surv
868	Pendiuk, J. E., Guarracino, L., Beich, M., Brunini, C. & Güntner, A (2020)
869	Estimating the specific yield of the pampeano aquifer argentina using super-
870	conducting gravimeter data. Hydrogeology Journal, 28(7), 2303–2313.
871	Permann, C. J., Gaston, D. R., Andrš, D., Carlsen, R. W., Kong, F., Lindsav, A. D.,

872	others (2020). Moose: Enabling massively parallel multiphysics simulation.
873	Software X, 11, 100430.
874	Qu, J., & Cherkaoui, M. (2006). Fundamentals of micromechanics of solids
875	(Vol. 735). Wiley Online Library.
876	Rau, G. C. (2018). PyGTide: A Python module and wrapper for ETERNA PRE-
877	DICT to compute synthetic model tides on Earth. Zenodo. Retrieved from
878	https://doi.org/10.5281/zenodo.1346260 doi: 10.5281/zenodo.1346260
879	Rau, G. C., Cuthbert, M. O., Acworth, R. I., & Blum, P. (2020). Disentangling the
880	groundwater response to earth and atmospheric tides to improve subsurface
881	characterisation. Hydrology and earth system sciences, $24(12)$ , $6033-6046$ .
882	Rau, G. C., Cuthbert, M. O., Post, V. E., Schweizer, D., Acworth, R. I., Andersen,
883	M. S., Ge, S. (2020). Future-proofing hydrogeology by revising groundwa-
884	ter monitoring practice. Hydrogeology Journal, 28(8), 2963–2969.
885	Roeloffs, E. A., Burford, S. S., Riley, F. S., & Records, A. W. (1989). Hydro-
886	logic effects on water level changes associated with episodic fault creep near
887	Parkfield, California. Journal of Geophysical Research, 94 (B9), 12387. Re-
888	trieved from http://doi.wiley.com/10.1029/JB094iB09p12387 doi:
889	10.1029/jb094ib09p12387
890	Rojstaczer, S. (1988). Determination of fluid flow properties from the response
891	of water levels in wells to atmospheric loading. Water Resources Research,
892	24 (11), 1927–1938.
893	Rojstaczer, S., & Agnew, D. C. (1989). The influence of formation material prop-
894	erties on the response of water levels in wells to earth tides and atmospheric
895	loading. Journal of Geophysical Research: Solid Earth, 94 (B9), 12403–12411.
896	Rojstaczer, S., & Riley, F. S. (1990). Response of the water level in a well to earth
897	tides and atmospheric loading under unconfined conditions. Water Resources $\mathcal{D}_{\text{construct}} = \mathcal{Q}(S)$ 1802–1817
898	Research, $20(6)$ , $1005-1017$ .
899	Schweizer, D., Ried, V., Rau, G. C., Tuck, J. E., & Stolca, P. (2021, 2). Compar-
900	Tidal Componenta from Croundwater Level Measurementa Mathemati
901	and Consciences Retrieved from http://link_springer_com/10_1007/
902	s11004-020-09915-9 doi: 10.1007/s11004-020-00015-0
903	Settari A & Walters D A (2001) Advances in coupled geomechanical and reser-
904	voir modeling with applications to reservoir compaction $Size Iournal 6(03)$
906	334-342.
907	Sun, X., Shi, Z., & Xiang, Y. (2020). Frequency dependence of in situ transmissivity
908	estimation of well-aquifer systems from periodic loadings. <i>Water Resources Re</i> -
909	search, $56(11)$ , e2020WR027536.
910	Turnadge, C., Crosbie, R. S., Barron, O., & Rau, G. C. (2019). Comparing meth-
911	ods of barometric efficiency characterization for specific storage estimation.
912	Groundwater, 57(6), 844–859.
913	Van Camp, M., & Vauterin, P. (2005, 6). Tsoft: graphical and interactive soft-
914	ware for the analysis of time series and Earth tides. $Computers \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \$
915	sciences, 31(5), 631-640. Retrieved from http://linkinghub.elsevier.com/
916	retrieve/pii/S0098300404002456 doi: 10.1016/j.cageo.2004.11.015
917	Verruijt, A. (2013). Theory and problems of poroelasticity. Delft University of Tech-
918	nology, 71.
919	Verruijt, A., & Van Baars, S. (2007). Soil mechanics. VSSD Delft, the Netherlands.
920	Virtanen, P., Gommers, R., Oliphant, T. E., Haberland, M., Reddy, T., Courna-
921	peau, D., others (2020). Scipy 1.0: fundamental algorithms for scientific
922	computing in python. Nature methods, $17(3)$ , $261-272$ .
923	Wang. (2017). Theory of linear poroelasticity with applications to geomechanics and
924	hydrogeology. Princeton University Press.
925	Wang, Doan, ML., Xue, L., & Barbour, A. J. (2018). Tidal response of ground-
926	water in a leaky aquifer—application to oklahoma. Water Resources Research,

927	54(10), 8019-8033.
928	Wang, & Manga, M. (2021). Response to tides, barometric pressure and seismic
929	waves. Water and Earthquakes, 83–153.
930	Wang, Zhu, AY., Liao, X., Manga, M., & Wang, LP. (2019). Capillary effects on
931	groundwater response to earth tides. Water Resources Research, $55(8)$ , $6886-$
932	6895.
933	Wang, C., & Zeng, Z. (2011). Overview of geomechanical properties of bakken
934	formation in williston basin, north dakota. In 45th us rock mechan-
935	$ics/geomechanics\ symposium.$
936	Wang, H. F. (2000). Theory of Linear Poroelasticity with Applications to Ge-
937	omechanics and Hydrogeology. Princeton University Press. Retrieved from
938	https://press.princeton.edu/titles/7006.html
939	Wenzel, HG. (1996). The nanoGal software: Earth tide data processing package:
940	Eterna 3.3. Bulletin d'Informations des Marées Terrestres, 124, 9425–9439.
941	Wilkins, A., Green, C. P., & Ennis-King, J. (2020). Porousflow: a multiphysics
942	simulation code for coupled problems in porous media. Journal of Open Source
943	Software, 5(55), 2176.
944	Wilkins, A., Green, C. P., & Ennis-King, J. (2021). An open-source multiphysics
945	simulation code for coupled problems in porous media. $Computers \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \$
946	sciences, 154, 104820.
947	Xue, L., Brodsky, E. E., Erskine, J., Fulton, P. M., & Carter, R. (2016). A perme-
948	ability and compliance contrast measured hydrogeologically on the san andreas
949	fault. Geochemistry, Geophysics, Geosystems, 17(3), 858–871.
950	Zhang, M., Hao, Y., Zhao, Z., Wang, T., & Yang, L. (2021). Estimation of coastal
951	aquifer properties: A review of the tidal method based on theoretical solutions.
952	Wiley Interdisciplinary Reviews: Water, 8(1), e1498.
953	Zhu, AY., & Wang. (2020). Response of leaky aquifers to earth tides–interpreted
954	with numerical simulation. Journal of Hydrology, 581, 124458.